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NuMI| Beam Status g i

* Relatively smooth running
« Occasional losses at “Hobbit Door”

* Thursday to thursday record week.
« 4.6 E18 — but we usually use Monday-Monday.

» Added 6™ batch for NuMI-only cycles.
 Increased target intensity limit to 2.9 E13
* EXxpect new target to be ready about 9/30.




Addition of 6" Batchto _&ms
NuMI-only Cycles

Page 3
B VMS GxPA 1: NuMI target hall

Man 19-SEFP-Z@860

E:TARTGT
aMumi Elg




Protons Delivered in Last
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Protons per Day since1 May = ="

Daily Protons Delivered to NuMI Target Starting 1 May 2005
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e Detector downtime small.

« HV problem at far detector gone since controller
noard replaced.

3 hour network outage in Minnesota on Tuesday.
« Detector ran fine during this period
« Have added upstairs monitoring connected to LAN.

Replaced 3 ND frontend electronic boards

« Still have afew noisy channels, and one at least that
will not trigger on cosmics.

Upgraded el ectronic logbook — thanks!
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Electronics mantenance at near detector
Far detector ~100% uptime.



